Matrices & Linear Algebra I C Term, Sections C01-C04 W. J. Martin February 18, 2002 Week 7: Reading and Exercises

Reading Previously, you were instructed to read Sections 6.6 and 6.7. Next, I should like you to carefully read Sections 1 and 2 of Chapter 8. Practice Exercises Note: Do not hand in.

Here are some practice problems to work on at home. If you have any questions about them, please discuss them in conference or come see me.

Previously assigned problems:

• page 292-3, # 1-11 (odd), 17, 19-21, 27-33 (odd), three from T.4-T.10. • page 303-5, # 1-25 (odd) (great problems!), T.1. Eigenvalue problems:

• page 354-5, # 1-21 (odd), T.9. • page 364-5, # 1-33 (odd), 37.